Author's response to reviews

Title: Papillary fibroelastoma of the left atrial wall: a case report

Authors:

Murat Bicer (mbicer23@yahoo.com)
Mustafa Cikirikcioglu (mustafa.cikirikcioglu@hcuge.ch)
Erman Pektok (erman.pektok@hcuge.ch)
Sarah Dettwiler (sarah.dettwiler@hcuge.ch)
Afksendiyos Kalangos (afksendyios.kalangos@hcuge.ch)
Hajo Muller (hajo.muller@hcuge.ch)

Version: 4 Date: 1 July 2009

Author's response to reviews:

Dear Editors,

As requested by the BioMed Central Editorial Production Team, please find revised text attached.

With my best regards,

Mustafa CIKIRIKCIOGLU, MD, PhD

Staff Surgeon and Research Correspondent (Chef de Clinique Scientifique)
Responsible of Minimal Invasive and Robotic Cardiac Surgery Programme

Division of Cardiovascular Surgery,
Department of Surgery University Hospital of Geneva
Rue Gabrielle-Perret-Gentil 4 - 1211 Genève 14

Phone (office) : (+41) 022 37 276 63
Phone (mobile) : (+41) 076 535 10 84
Pager : 68 57 651 (for external calls (+41) 022 37 23 311)
Fax : (+41) 022 37 276 34

e-mail : mustafa.cikirikcioglu@hcuge.ch